Author's response to reviews

Title: A predictive model for the early identification of patients at risk for a prolonged intensive care unit length of stay

Authors:

Andrew A Kramer (akramer@cerner.com)
Jack E Zimmerman (jackcernerapache@yahoo.com)

Version: 4 Date: 3 May 2010

Author's response to reviews:

No response needed.